

## **Research Article**

Check for updates

# Learning together for mental health: feasibility of measures to assess a whole-school mental health and wellbeing intervention in secondary schools

Oliver Lloyd-Houldey<sup>®</sup>,<sup>1</sup> Joanna Sturgess<sup>®</sup>,<sup>2</sup> Neisha Sundaram<sup>®</sup>,<sup>3</sup> Steven Hope<sup>®</sup>,<sup>1</sup> Semina Michalopoulou<sup>®</sup>,<sup>1</sup> Elizabeth Allen<sup>®</sup>,<sup>2</sup> Lee Hudson<sup>®</sup>,<sup>1</sup> Stephen Scott<sup>®</sup>,<sup>4</sup> Dasha Nicholls<sup>®</sup>,<sup>5</sup> Deborah Christie<sup>®</sup>,<sup>6</sup> Rosa Legood<sup>®</sup>,<sup>7</sup> Chris Bonell<sup>®</sup><sup>3</sup> and Russell Viner<sup>®</sup><sup>1\*</sup>

<sup>1</sup>Population Policy and Practice Research and Teaching Department, University College London Great Ormond Street Institute of Child Health, London, UK

<sup>2</sup>Department of Medical Statistics, London School of Hygiene & Tropical Medicine, London, UK

<sup>3</sup>Department of Public Health, Environments and Society, London School of Hygiene & Tropical Medicine, London, UK <sup>4</sup>Institute of Psychiatry, Psychology & Neuroscience, King's College London, London, UK

<sup>5</sup>Division of Psychiatry, Imperial College London, London, UK

<sup>6</sup>Department of Behavioural Science and Health, University College London, Institute of Epidemiology and Health Care, London, UK

<sup>7</sup>Department of Health Services Research and Policy, London School of Hygiene & Tropical Medicine, London, UK

### \*Corresponding author r.viner@ucl.ac.uk

Published June 2025 DOI: 10.3310/GFDT2323

## Abstract

**Background:** Population mental health in young people worsened during and since the COVID-19 pandemic. School environments can play a key role in improving young people's mental health. Learning Together for Mental Health is a whole-school intervention aiming to promote mental health and well-being among young people in secondary schools. Before progressing to a Phase III effectiveness evaluation of the intervention, it is critical to assess the feasibility of trial measures at baseline and follow-up.

**Objective:** To evaluate the feasibility of trial measures and procedures within a feasibility study of a whole-school intervention aiming to promote mental health and well-being among young people in secondary schools, including whether we met our progression criterion of survey response rates of 60% or more in two or more schools at baseline and follow-up.

**Design and methods:** We conducted a feasibility study which included assessment of the indicative primary and secondary outcomes measures and procedures to be used in a future Phase III trial.

**Setting and participants:** Setting for our feasibility study included five state, mixed-sex secondary schools in southern England (one of which dropped out after baselines and one of which replaced this). We recruited year-7 students to participate in the baseline survey and year-10 students to participate in the follow-up survey at 12-month follow-up. Baseline and follow-up participants were different groups, as the focus was assessing feasibility of measures for the age groups that would be surveyed at baseline and follow-up in a Phase III randomised controlled trial. Our study was not powered or designed to estimate intervention effects.

**Interventions:** As part of our feasibility study, all schools received the Learning Together for Mental Health intervention for one academic school year.

**Main outcome measures:** The indicative primary outcome measure trialled was the total difficulties score of the Strengths and Difficulties Questionnaire. Indicative secondary outcomes measures trialled were the: Warwick-Edinburgh Mental Well-being Scale; Short Moods and Feelings Questionnaire; Generalised Anxiety Disorder-7 scale;

Eating Disorders Examination – Questionnaire Short, self-harm (single item from the Health Behaviour in Schoolaged Children study); bullying victimisation (Gatehouse Bullying Scale); cyberbullying (two items adapted from the Dose Adjustment for Normal Eating II questionnaire); substance use (National Health Service measure); and Beyond Blue School Climate Questionnaire.

**Results:** Trial measures and procedures were feasible to implement and were acceptable to year-7 and year-10 students, teachers and parents. At baseline, response rates ranged from 58% to 91% between schools. Only two students were opted out by parents, and no students opted out in advance. Students refusing consent on the day of survey was rare (7%). Twelve per cent of students were absent. The follow-up survey had an overall response rate of 66%, ranging from 44% to 91%. Only two students were opted out by parents, and three students opted out in advance. Overall, 12% opted out on the day. Twenty per cent of students were absent. Variation in response rate reflected specific problems at certain schools. Surveys took 40–45 minutes at baseline and 30 minutes at follow-up. The trial progression criterion concerning response rates was achieved, with three of four schools at baseline and two of four schools at follow-up having responses rates above 60%.

**Limitations:** Our study involved a small, purposive sample of schools and students which are not representative of those in England.

**Conclusions:** With some minor amendments, trial measures and procedures should be applied in a future Phase III effectiveness evaluation of the Learning Together for Mental Health intervention.

**Future work:** Survey response rates could be improved if baseline and follow-up surveys are not scheduled in the last weeks of term, on Fridays or near mock General Certificate of Secondary Education exams. Completion of some measures (such as Eating Disorders Examination – Questionnaire Short) among year-7 students may be improved if question wording is tailored to be age-appropriate.

**Funding:** This article presents independent research funded by the National Institute for Health and Care Research (NIHR) Public Health Research programme as award number NIHR131594.

A plain language summary of this research article is available on the NIHR Journals Library Website https://doi. org/10.3310/GFDT2323.

## Introduction

Mental disorders are the predominant cause of disability in the UK,<sup>1</sup> with onset by age 14 in approximately 50% of cases, and onset by age 24 in approximately 75% of cases.<sup>2,3</sup> Approximately 20% of young people aged 8-25 years in England have one or more mental disorder.<sup>4</sup> Disordered eating is present in approximately 40% of adolescent girls in the UK,5 and 26% of women aged 16-24 in England report having ever self-harmed.<sup>6</sup> The green paper 'Transforming Children and Young People's Mental Health Provision' identified a prominent role for schools in England and Wales in promoting mental health (MH).7 Historically, schools have generally not been supported by specialist MH expertise. As of spring 2023, the NHS England and Department for Education's joint roll-out of MH support teams was covering 35% of pupils in schools and learners in further education in England.<sup>8</sup> This programme, however, is still focused predominantly on responding to MH problems, as opposed to prevention, and there remains a need to develop and evaluate new universal preventive interventions.9 In light of recent evidence indicating that young people's MH has worsened since the COVID-19 pandemic, developing effective, evidence-based universal interventions is of even greater importance.<sup>10,11</sup>

Evidence from multiple reviews suggests there is potential for school programmes to address young people's disruptive behaviours and MH,<sup>12-15</sup> including body image and disordered eating;<sup>16</sup> anxiety and depression;<sup>13</sup> selfharm; and development of supportive capacities, for example, self-regulation.<sup>17</sup> However, there are significant limitations to these programmes and the existing evidence base, including small effect sizes,<sup>12</sup> interventions addressing single facets of MH;<sup>18</sup> insufficient flexibility in implementation and delivery;<sup>19</sup> interventions not proving effective at scale; and a lack of interventions co-designed with young people.<sup>9</sup>

Whole-school interventions, which go beyond classroom curricula to encompass modifications of the wider school environment, have been shown to be effective for promoting physical and MH outcomes and are acceptable to schools.<sup>9,20,21</sup> However, their application to the field of MH has been limited, and, therefore, there is a lack of evidence to support their implementation.<sup>20</sup> This type of intervention aims to address multiple determinants of MH via the school environment, including school culture and systems.<sup>22</sup> Such interventions also aim to increase student engagement with the school, particularly among the most disadvantaged,<sup>23,24</sup> as well as build ownership among school leaders, staff and students. School observational studies suggest that the school environment can influence student MH in various ways, including through: development of social and emotional skills, influencing school belonging and commitment, engagement with prosocial or antisocial

2

peers, experience of bullying and involvement in social-support networks.  $^{\rm 25\mathchar`25\mathchar`25\mathchar`28\mathchar`25\mathchar`25\mathchar`25\mathchar`28\mathchar`25\mathchar`28\mathchar`25\mathchar`28\mathchar`28\mathchar`28\mathchar`28\mathchar`28\mathchar`28\mathchar`28\mathchar`28\mathchar`28\mathchar`28\mathchar`28\mathchar`28\mathchar`28\mathchar`28\mathchar`25\mathchar`28\mathcha$ 

We previously conducted the INCLUSIVE cluster randomised trial of the Learning Together (LT) intervention with 40 secondary schools.<sup>29</sup> LT incorporated a whole-school approach, aiming to influence the school environment and reduce bullying via: assessment of student needs; a student/staff action group (with external facilitator support) tasked with reviewing needs data and using this to decide on and implement changes to school policies and the wider school environment; training for school staff to use restorative practice (RP) to intervene in bullying and other conflict; and a social and emotional learning (SEL) curriculum. With the exception of the curriculum, we found all of these were implemented with fidelity. The intervention had significant beneficial effects, reducing bullying victimisation (primary outcome) in addition to improving mental well-being and health-related quality of life, and reducing psychological distress and substance use (secondary outcomes), with high-cost effectiveness.<sup>29</sup> Although the intervention did not include activities other than the curriculum directly addressing MH, effect sizes on MH outcomes were approximately 0.1 standard deviation (SD).

In light of these findings, we undertook a refinement and feasibility study, aimed at modifying LT to focus on MH and well-being. We assessed the feasibility and acceptability of the modified intervention, Learning Together for Mental Health (LTMH),<sup>30</sup> in secondary schools in England to determine progression to a potential Phase III trial. The LTMH intervention involved the following components:

- 1. **Mental health needs-assessment report** informed by survey of year-7 students.
- 2. Action group comprising around six staff and around six students, supported by an external facilitator, conducting at least six meetings to review the needs report, identify changes to school policies and environment, select actions from a menu of evidence-based options to address student MH needs, and oversee implementation of the programme.
- 3. **Restorative practice** implemented in response to student bullying and other conflict, with all teaching staff receiving introductory training on RP and four to five selected staff receiving intensive training on RP.
- 4. **Bounce Forward Healthy Minds (SEL) curriculum**<sup>31,32</sup> with selected teaching staff receiving training from the Bounce Forward charity to deliver the 'Foundation Resilience Skills' module to all year-8 students and other modules should the action group members and school leaders decide this would be useful.

Learning Together for Mental Health was implemented in the 2022–3 school year in four mixed-sex, state secondary schools along with a feasibility study and process evaluation. Student data on outcomes we identified as the primary and secondary indicative outcomes for a future full trial were assessed using paperbased questionnaires at baseline among year-7 students and at 12-month follow-up among year-10 students. These data were summarised for schools to provide the MH needs-assessments report.

In this paper, we report on findings from the baseline and follow-up surveys and examine the feasibility and acceptability of the indicative outcome measures and operating procedures to be used in a future trial.

## **Aim and objectives**

We aimed to assess the feasibility and acceptability of indicative outcomes measures, as well as trial and survey operating procedures (e.g. determining response rates) at baseline and follow-up to inform a potential future Phase III trial.

## Methods

A full description of the feasibility study and process evaluation, including patient and public involvement and engagement (PPIE) are reported elsewhere.<sup>33,34</sup>

## Design

We aimed to conduct a pilot intervention and feasibility study among a purposive sample of four state secondary schools in southern England, incorporating a process evaluation, as well as baseline and follow-up surveys to assess the feasibility and acceptability of outcome measures, and trial and survey operating procedures. This involved piloting all primary and secondary outcome measures (described in this paper as indicative outcome measures), with year-7 students at baseline and year-10 students at 12-month follow-up, as well as employing all standard operating procedures (SOPs) for data collection and obtaining consent as would be applied in a Phase III cluster randomised trial. Full details of the study, including the study population recruitment methods and participant flow, are described elsewhere.

## Fieldwork and data collection

Paper questionnaires containing the MH and other measures were completed confidentially by students in classrooms under exam conditions (see *Report* 

Lloyd-Houldey O, Sturgess J, Sundaram N, Hope S, Michalopoulou S, Allen E, et al. Learning together for mental health: feasibility of measures to assess a whole-school mental health and wellbeing intervention in secondary schools [published online ahead of print June 25 2025]. Public Health Res 2025. https://doi.org/10.3310/GFDT2323

Supplementary Material 1, supplementary files 1 and 2). One or two fieldworkers were allocated per classroom, where they were responsible for distributing consent materials (see Report Supplementary Material 1, supplementary files 3 and 4) and questionnaires, explaining the study, managing consent, referring any safeguarding issues to the trial manager, supporting students with questionnaire completion, collecting completed questionnaires and completing a data collection record for their class. The trial manager and one study researcher attended all fieldwork sessions at schools, and supported fieldworkers as required. Teachers remained at the front of the class to maintain quiet and order but were unable to read students' responses. Fieldworkers, and, in some cases, teaching assistants, supported students with mild learning difficulties or low English proficiency. Where teaching assistants were involved, they read questions and response options aloud but did not see student responses. During the baseline survey, questionnaires and consent forms for absent students were left with the school study implementation leads for students to complete on their return along with stamped, addressed envelopes for schools to return the completed materials. This did not result in any returned questionnaires and so we did not repeat this procedure at follow-up.

All fieldworkers received online training and were equipped with student information sheets, consent forms, data collection record, teacher information sheets, fieldwork SOPs, questionnaires and frequently asked questions about the questionnaire (see *Report Supplementary Material* 1, supplementary files 1–8).

#### **Outcome measures**

The primary outcome for the overall study was meeting progression criteria to proceed to a Phase III trial of effectiveness. Those concerning intervention feasibility and acceptability are reported elsewhere.<sup>33,34</sup> In this paper, we report on the criterion of whether we achieved a survey response rate of 60% or more in two or more schools at baseline and follow-up, as well as reporting on the feasibility and acceptability of indicative outcomes measures, described below.

#### Indicative primary outcome

The Strengths and Difficulties Questionnaire (SDQ)<sup>35</sup> total score was chosen as the indicative primary outcome for a future trial. The SDQ is the most widely used MH outcome measure for children and young people in the UK, extensively validated in population-samples with children of secondary-school age and used in recent national MH surveys in England.<sup>36,37</sup>

#### Indicative secondary outcomes

Indicative secondary outcomes in a future trial were assessed, including:

- SDQ subscales: emotional problems, conduct problems, peer problems, hyperactivity problems and the prosocial subscale.<sup>35</sup>
- Well-being: Warwick-Edinburgh Mental Well-being Scale<sup>38</sup>
- 3. Depressive symptoms: Short Moods and Feelings Questionnaire (SMFQ)<sup>39</sup>
- 4. Anxiety, measured using the seven-item Generalised Anxiety Disorder-7 (GAD-7) scale<sup>40</sup>
- Disordered eating behaviour and cognitions, measured using the Eating Disorders Examination – Questionnaire Short (EDE-QS), which has measures of weight and shape concerns<sup>41</sup>
- Self-harm: using one question derived from the Health Behaviour in School-aged Children (HBSC) study<sup>42</sup>
- Bullying (victimisation) measured using the Gatehouse Bullying Scale (GBS)<sup>43</sup>
- Cyberbullying, assessed using two items adapted from the DAPHNE II questionnaire asking whether the participant was bullied (victim) and/or bullied someone else (perpetrator) through mobile phone use or the internet<sup>44</sup>
- 9. Substance use (NHS measure)<sup>45</sup>
- 10. Student report of school climate, using the Beyond Blue School Climate Questionnaire (BBSCQ)<sup>46</sup>

We also piloted the Child Health Utility – 9 Dimension measure at baseline and follow-up required for an economic analysis within a future Phase III trial of effectiveness, and this is reported elsewhere.

#### Statistical analyses

The analyses reported here aimed to determine the feasibility and acceptability of our use of outcome measures and trial procedures. We report descriptive summaries of baseline and follow-up data, assessing missing data. We also assess the reliability of outcome measures using Cronbach's alpha where appropriate.

All deviations from the feasibility study protocol are outlined elsewhere.

#### Ethics, safeguarding and data protection

Ethics approval for baseline and follow-up surveys was obtained from University College London (UCL) Research Ethics Committee (REC) on 30 March 2022 (UCL Ethics Project ID Number: 21179/001) and the London School of Hygiene and Tropical Medicine (LSHTM) REC on 26 August 2022 (ref. 27994). The trial manager, who led fieldwork teams during surveys, had an enhanced Disclosure and Barring Services (DBS) check so he could work unaccompanied in schools. We originally planned that all fieldworkers visiting schools would have a DBS check, but this was not required since fieldworkers did not work unsupervised with students. All activities were carried out in accordance with guidelines outlined by the Economic and Social Research Council, the Data Protection Act 1998, the latest Directive on Good Clinical Practice (2005/28/EC) and the General Data Protection Regulation 2018.

Headteachers were asked for written informed consent for the intervention (see Report Supplementary Material 1, supplementary file 9). Informed written opt-in consent was sought from students judged competent to provide this. Students and their parents were sent an information sheet (see Report Supplementary Material 1, supplementary files 3 and 10) several days before data collection. These explained that participation was voluntary and withdrawal was possible at any point with no negative consequences. Information sheets also provided contact details of the research team should participants or parents have questions. Students or their parents could opt out of the research in advance by contacting the school or research team. Just before data collection, students received an oral description of the study and could ask questions before signing a consent form (see Report Supplementary Material 1, supplementary file 4). They were advised that participation was voluntary, and they could withdraw at any point or skip any question they did not want to answer.

Students were advised that their responses would be treated as confidential, but if they reported or disclosed to researchers any indication of risk of serious harm, anonymity would be broken, and their name and details of the potential risk shared with the school's safeguarding lead to decide what action was required. Students opting out of the survey were advised to stay in the classroom and complete reading, homework or private study. We used SOPs for dealing with safeguarding concerns (see Report Supplementary Material 1, supplementary file 11) and reporting serious adverse events (SAEs) (see Report Supplementary Material 1, supplementary file 12). We balanced our ethical duties of promoting participant autonomy by respecting confidentiality and of promoting participant well-being when determining when we would need to breach confidentiality to address abuse. When such abuse was reported through a questionnaire, we contacted the school safeguarding lead. When it was reported to research staff, we discussed this with the participant prior to contacting the safeguarding lead.

Survey data were managed by research staff using secure systems. Self-report data linked to participant unique identification codes were stored by LSHTM clinical trials unit. Student names linked to the same identification codes were stored by the UCL fieldwork team. Both used password-protected drives and folders. In line with Medical Research Council (MRC) guidance on personal information in medical research, we will retain survey data for 20 years after study completion. No single institution held self-report data linked to names. But where the former indicated potential safeguarding issues, self-report data could be linked to name by interinstitutional collaboration.

Throughout the study, we requested information from participating schools on SAEs among students and assessed whether these were plausibly related to the intervention or research in consultation with school leads. The study steering committee and LSHTM REC were provided with anonymised reports of all safeguarding referrals and SAEs.

## Results

School recruitment and participant flow are reported in full elsewhere. Overall, four schools were recruited to the study and completed baseline surveys. Study schools were slightly lower on free school meal (FSM) entitlement and higher on government inspection rating compared to schools in all of England. In these, 640 (79%) of year-7 students completed baseline surveys. One school dropped out of the study immediately after baseline surveys in September 2022 because of challenges with workloads and was replaced by another school in November 2022 which had previously expressed interest. A baseline survey was not completed in this school. In the four schools remaining in the study, 566 (66%) of year-10 students completed follow-up surveys. School characteristics compared to schools in England are reported elsewhere.

#### Baseline student surveys

Baseline year-7 surveys were conducted between 12 July 2022 and 11 October 2022. Baseline surveys with study schools 2, 3 and 4 were completed during July 2022. Rail strikes delayed travel to study school 1, and consequently, only half of students were surveyed during the first visit in July 2022. The remaining half were surveyed on 11 October 2022 when they had started year 8.

Response rates ranged from 58% to 91% between schools (*Table 1*). Response rates were lowest at school 2 (80%) and school 4 (58%). No students were identified by schools as not competent to give consent. Only two students were opted out by parents, and no students opted out in advance. Students refusing consent on the day of survey was rare overall (7%). School 4 had a relative high proportion of students refusing consent on the day (17%) which contributed to its low response rate (58%). Twelve per cent of students were absent on the day.

Sixty-seven students submitted incomplete consent forms but completed questionnaires. Review of these consent forms in consultation with UCL REC confirmed that there was sufficient evidence that 54 of these were consenting to participate in the research. Some students failed to write their full name or date in the consent form, or did not understand the requirement to tick all 15 consent box statements. Data from these 54 students were included in the analysis. Thirteen (2%) students were deemed to have given insufficient consent, as they had either not signed the consent form or had marked consent boxes with a mixture of crosses and ticks: subsequently, their questionnaires were destroyed and data not analysed. Overall, 12% of students were absent on the day of surveys, and none of these returned the questionnaires left for them. Absentees were highest at school 1 (15%) and 4 (22%).

Questionnaires took students 40–45 minutes to complete, in line with our previous research.<sup>29</sup> There were no concerns or complaints raised about the questionnaire by students, teachers or parents. Student queries focused on clarification of language and meaning, which were answered by fieldworkers. Terms sometimes not understood included 'markedly' (as used in the Likert scale included in the EDE-QS), and 'full time' and 'part time' in

relation to a question on parental work. Some students queried the purpose of the questions included in the Family Affluence Scale (FAS). Some students were unsure how to describe their ethnicity. Some students were uncertain about the sequencing of questions for the SDQ impact supplement. Fieldworkers were able to address these issues. Safeguarding issue identified during surveys are reported below.

#### Follow-up student survey

Fieldwork procedures remained largely unchanged from baseline surveys. Minor changes were: having one fieldworker only per class but with the trial manager and study researcher supporting across classes; not leaving surveys for absent students; in agreement with our ethics committees, updating the student consent form with one tick box for all statements of consent. Procedures had to be adapted at study school 5 due to significant challenges described below.

Follow-up surveys were conducted between 15 June 2023 and 17 July 2023. Surveys at schools 1 and 3 were completed in June, but surveys at schools 4 and 5 had to be within the last week of term due to school timetabling challenges. We experienced significant challenges surveying at school 5 due to organisation and timetabling issues on the day. Around half of the students were called out of their classrooms near the start of the survey to attend a year-10 activity which had not been communicated to the study team and were gone for all or most of the session. Most had not yet consented or started the questionnaire and were recorded as absent. Of these, only 33 students completed the survey later at school.

The majority of students completed the questionnaire in approximately 30 minutes. There were no concerns or

	Feasibility study/intervention schools					
Baseline survey	1	2	3	4	5	Total
Ineligible, n (% eligible)	0/216 (0)	0/211 (0)	0/210 (0)	0/171 (0)	N/A	0/808 (0)
Parent opt-out ahead, n (% eligible)	2/216 (1)	1/211 (1)	3/210 (1)	0/171 (0)	N/A	6/808 (1)
Student opt-out ahead, n (% eligible)	0/216 (0)	0/211 (0)	0/210 (0)	0/171 (0)	N/A	0/808 (0)
Absent on the day, <i>n</i> (% eligible)	33/216 (15)	23/211 (11)	2/210 (1)	38/171 (22)	N/A	96/808 (12)
Student opt-out on day, n (% eligible)	2/216 (1)	13/211 (6)	9/210 (4)	29/171 (17)	N/A	53/808 (7)
Insufficient consent (data removed from analysis), n/eligible	0/216 (0)	5/211 (2)	4/210 (2)	4/171 (2)	N/A	13/808 (2)
Overall response rate, n/eligible (%)	179/216 (83)	169/211 (80)	192/210 (91)	100/171 (58)	N/A	640/808 (79)

#### **TABLE 1** Survey response rates at baseline

complaints raised about the questionnaire by students, teachers or parents, and students asked very few questions concerning the questionnaire. Guidance concerning the sequencing of questions in the SDQ impact supplement was updated, and no students raised issues about this. No students exhibited distress, and no safeguarding concerns were identified.

The follow-up survey had an overall response rate of 66%, ranging from 44% to 91% (*Table 2*), reflecting a timetabling clash with a school trip and high rates of absenteeism on the day in school 4, the challenges reported above at school 5, and high opt-out rates (12% and 27%) in schools 4 and 5.

Only two students were opted out by parents, and three students opted out in advance. Overall, 12% of students opted out on the day with significant variation between schools. One student in school 4 was deemed not competent to give consent. Twenty per cent of students were absent.

All data management procedures were feasible, including the use of self-report data linked to unique identification codes.

#### Safeguarding

Safeguarding concerns were raised for four students during baseline surveys: one was upset and spoke of homophobic bullying they had experienced; two were not distressed but spoke of bullying and self-harming that they had experienced; and one became very distressed, worrying that their responses would be shared with their social worker, which would result in the student being taken into care. Analysis of handwritten responses on baseline questionnaires identified safeguarding concerns relating to two other students who reported self-harm. All these students were referred to school safeguarding leads. Analysis of handwritten responses on follow-up questionnaires identified safeguarding concerns for one student who reported the impact of bullying that they had experienced, and who was referred to the school safeguarding lead.

#### Serious adverse events

No SAEs were reported by study schools.

#### **Student characteristics**

Student sociodemographic characteristics are shown in *Table 3*. We did not compare differences in sociodemographic data between baseline and follow-up as the surveys were conducted on different groups of students.

In general, most students who participated in the year-7 baseline survey and year-10 follow-up survey completed the sociodemographic items, and rates of missing data were low. With the exception of the FAS, sociodemographic items had < 10% missing data at baseline or follow-up. Missing data were higher for questions on family affluence, family structure, sexual orientation and ethnicity than other measures. Missing data were higher at baseline than follow-up across all sociodemographic survey items.

The Cronbach's alpha value for the FAS was lower (0.64) at baseline than follow-up (0.72). There were slightly more male year-7 students at baseline (51%) and slightly more female year-10 students at follow-up (54%). The two samples differed in terms of gender, sexual

#### TABLE 2 Survey response rates at follow-up

	Feasibility stu					
Follow-up survey	1	2	3	4	5	Total
Ineligible, n (% eligible)	0/213 (0)	N/A	0/180 (0)	1/211 (1)	0/254 (0)	1/858 (0)
Parent opt-out ahead, n (% eligible)	0/213 (0)	N/A	0/180 (0)	0/211 (0)	2/254 (1)	2/858 (0)
Student opt-out ahead, n (% eligible)	1/213 (1)	N/A	1/180 (1)	0/211 (0)	1/254 (0)	3/858 (0)
Absent on the day, <i>n</i> (% eligible)	12/213 (6)	N/A	8/180 (4)	87/211 (41)	62/254 (24)	169/858 (20)
Student opt-out on day, <i>n</i> (% eligible)	7/213 (3)	N/A	6/180 (3)	26/211 (12)	68/254 (27)	107/858 (13)
Insufficient consent – data removed from analysis, n/eligible	0/213 (0)	N/A	1/180 (1)	0/0 (0)	9/254 (4)	10/858 (1)
Overall response rate, n/eligible (%)	193/213 (91)	N/A	164/180 (91)	97/211 (46)	112/254 (44)	566/858 (66)
N/A not applicable						

This article should be referenced as follows

Lloyd-Houldey O, Sturgess J, Sundaram N, Hope S, Michalopoulou S, Allen E, et al. Learning together for mental health: feasibility of measures to assess a whole-school mental health and wellbeing intervention in secondary schools [published online ahead of print June 25 2025]. Public Health Res 2025. https://doi.org/10.3310/GFDT2323

## TABLE 3 Year-7 and year-10 student characteristics

	Year-7 (baseline) surveyª			Year-10 (follow-up) survey <sup>a</sup>			
Student characteristic	Number or mean	% or SD	Cronbach's alpha	Number or mean	% or SD	Cronbach's alpha	
Total n	640	100%		566	100%		
Age	11.9	0.36		14.8	0.40		
Missing	9	1.4%		0	0%		
Sex							
Male	328	51.2%		259	45.8%		
Female	306	47.8%		306	54.1%		
Missing	6	0.9%		1	0.2%		
Gender							
Воу	327	51.1%		252	44.5%		
Girl	292	45.6%		301	53.2%		
Non-binary	3	0.5%		4	0.7%		
Other	8	1.3%		7	1.2%		
Missing	10	1.6%		2	0.4%		
Sexual orientation							
Straight or heterosexual	514	80.3%		515	91.0%		
Gay or lesbian	5	0.8%		8	1.4%		
Bisexual	26	4.1%		18	3.2%		
Asexual	9	1.4%		2	0.4%		
Unsure/questioning	37	5.8%		13	2.3%		
Other	10	1.6%		2	0.4%		
Missing	39	6.1%		8	1.4%		
Ethnicity							
White	268	41.9%		149	26.3%		
Asian/Asian British	241	37.7%		294	51.9%		

NIHR Journals Library www.journalslibrary.nihr.ac.uk

## TABLE 3 Year-7 and year-10 student characteristics (continued)

	Year-7 (baseline) survey <sup>a</sup>			Year-10 (follow-up) survey <sup>a</sup>			
Student characteristic	Number or mean	% or SD	Cronbach's alpha	Number or mean	% or SD	Cronbach's alpha	
Black/Black British	42	6.6%		42	7.4%		
Mixed ethnicity	33	5.2%		32	5.7%		
Other	16	2.5%		44	7.8%		
Missing	40	6.3%		5	0.9%		
Family structure							
Two parents	488	76.3%		430	76.0%		
Single mother	60	9.4%		55	9.7%		
Single father	5	0.8%		3	0.5%		
Reconstituted	25	3.9%		19	3.4%		
Other	6	0.9%		12	2.1%		
Missing	56	8.8%		47	8.3%		
Parent/guardian in paid work							
Yes	519	81.1%		487	86.0%		
No	24	3.8%		29	5.1%		
Don't know	76	11.9%		39	6.9%		
Missing	21	3.3%		11	1.9%		
FAS							
Total score	8.4	2.35	0.64	8.5	2.68	0.72	
Missing	107	16.7%		18	3.2%		

a Due to study school 2 dropping out of the study after the baseline survey, schools included in the baseline survey are different to schools included in the follow-up survey. Schools surveyed in the baseline are study schools 1, 2, 3 and 4. Schools surveyed at follow-up are study schools 1, 3, 4 and 5.

orientation, ethnicity, parental working and FAS but not family structure.

#### Indicative outcome measures

The majority of indicative outcome measures had < 10% missing data at baseline or follow-up (*Table 4*), with most students participating in the survey completing items included under each measure. Items or scales which reported missing data above 10% were self-harm, GBS, SMFQ and Short Warwick-Edinburgh Mental Well-being Scale (SWEMWBS). For all these scales, missing data were higher among year-7 than year-10 students. The study's indicative primary outcome measure, the total difficulty score of the SDQ, was well completed, with only approximately 4% students not completing this scale in baseline or follow-up surveys.

Generally, scales demonstrated good inter-item reliability, with the majority having Cronbach's alpha values of 0.8 or higher (a value at or above 0.7 is generally regarded as indicating good inter-item reliability). GBS and the SDQ subscales were the only scales with a Cronbach's alpha value lower than 0.7. The total difficulty score for the SDQ had a Cronbach's alpha value of approximately 0.8 at baseline and 0.7 at follow-up, suggesting reasonably good inter-item reliability.

## Discussion

#### Summary of key findings

Trial and survey operating procedures, as well as indicative outcome measures, were feasible to implement and acceptable to year-7 and year-10 students, teachers and parents. We achieved the relevant progression criterion in terms of three of four schools at baseline and two of four schools at follow-up having responses rates above 60%. With the exception of challenges identified with the year-7 consent form, consent procedures in general worked well. This approach to consent is in line with MRC guidance<sup>47</sup> and common law.<sup>48</sup> It adds to a growing evidence base, indicating an opt-out parental consent approach is both feasible and acceptable to students and schools.<sup>26,46</sup> The revised consent form used during the follow-up survey performed better than the one used for the baseline survey. As found with other school-based survey research,<sup>29,49</sup> the use of paper-based questionnaire surveys was feasible to implement and appeared acceptable to students and teachers. Minimal distress was observed among year-7 students at baseline and no distress observed among year-10 students at follow-up.

In general, survey instruments performed well at baseline and follow-up. Proportions of missing data were generally low, including for the indicative primary outcome measure. Scales with missing data of 10% or higher at baseline included the FAS, SWEMWBS, EDE-QS, GBS and SMFQ. Fieldworker notes indicate that some year-7 students were confused by some of the language in the FAS and EDE-QS scales. Scales generally demonstrated good inter-item reliability.

#### Limitations

The feasibility study involved a small, purposive sample of schools and students which are not representative of those in England. We cannot assume that trial procedures would be as feasible in other schools, particularly those with lower inspection ratings or higher rates of FSM eligibility. The study did involve PPIE reported elsewhere, but this did not focus on measures or trial procedures. The study did not aim to assess effectiveness and did not follow a cohort of students from baseline to follow-up.

#### Implications for research and policy

Rail strikes delayed some baseline data collection at one school. Timetabling issues impeded follow-up surveys at another school. These and other challenges which resulted in low response rates in some schools can be avoided in a future Phase III trial by conducting surveys earlier in the school year. Surveys held towards the end of the school year in July tended to have higher rates of absenteeism and opting out among students. In particular, the baseline survey at school 4 and follow-up surveys at schools 4 and 5 were all held in the last five days of term in late July with 22%, 41% and 24% of students absent from class, respectively, at the time of these surveys. Year-10 students surveyed in the last 5 days of term at schools 4 and 5 were generally unsettled, with relative higher rates of students opting out of the survey. Attendance data from primary and secondary schools in England indicates absenteeism is highest on Fridays and in the last week of the autumn term;<sup>50-52</sup> however, attendance has been worse generally since the COVID-19 pandemic and has not yet fully returned to pre-pandemic levels.<sup>52</sup> Scheduling surveys for year-10 students was particularly challenging due to busy end-of-term schedules and further complicated at school 4 due to mock General Certificate of Secondary Education (GCSE) exams being held. Response rates could be improved if baseline and follow-up surveys were not scheduled in the last weeks of term, on Fridays or near mock GCSE exams. Our revised consent form used with year-10 students with simpler box ticking should be used for all students in any future Phase III trial. Where there were challenges with student understanding and completion of some items, such as FAS and EDE-QS, it may be useful to refine question wording and to use cognitive testing and piloting to assess the impact of this. Data management procedures, including the use of self-report data linked to

		Year-7 (baseline) survey			Year-10 (follow-up) survey		
Indicative outcome measures		Number or mean	% or SD	Cronbach's alpha	Number or mean	% or SD	Cronbach's alpha
SDQ	'High' level of total problems (%)		24.3			20.5	
	Total score (mean, SD)	12.9	6.7	0.83	12.6	5.9	0.73
	Missing total score (n, %)	23	3.6		20	3.5	
	Prosocial subcale (mean, SD)	7.8	1.9	0.70	7.2	1.9	0.64
	Emotional subscale (mean, SD)	4.0	2.7	0.75	3.9	2.7	0.76
	Conduct subscale (mean, SD)	2.1	1.9	0.59	2.0	1.7	0.55
	Hyperactivity subscale (mean, SD)	4.7	2.7	0.78	4.7	2.5	0.74
	Peer subscale (mean, SD)	2.1	1.9	0.63	2.0	1.7	0.60
	Impact score (mean, SD)	1.1	1.9	0.78	1.0	1.8	0.77
Well-being: SWEMWBS	Total score (mean, SD)	22.0	5.1	0.85	21.8	5.2	0.88
	Missing (n, %)	77	12.0		48	8.5	
Depressive symptoms: SMFQ	Total score (mean, SD)	6.4	6.6	0.93	6.2	6.3	0.93
	High score – no (n, %)	451	70.5		414	73.1	
	High score - yes (n, %)	117	18.3		102	18.0	
	Missing (n, %)	72	11.3		50	8.8	
Anxiety: GAD-7 scale	Total score (mean, SD)	5.9	6.0	0.92	5.7	5.8	0.92
	High score – no (n, %)	447	69.8		401	70.8	
	High score – yes (n, %)	148	23.1		124	21.9	
	Missing	45	7.0		41	7.2	
Eating behaviour and cognitions: EDE-QS	Total score (mean, SD)	7.8	8.0	0.90	8.1	8.3	0.91
	High score – no (n, %)	465	72.7		407	71.9	
	High score – yes (n, %)	113	17.7		110	19.4	
	Missing (n, %)	62	9.7		49	8.7	
							continued

Public Health Research 2025

 TABLE 4
 Indicative primary and secondary outcome measures at baseline (year-7 students) and follow-up (year-10 students) (continued)

		Year-7 (baseline) survey			Year-10 (follow-up) survey		
Indicative outcome measures		Number or mean	% or SD	Cronbach's alpha	Number or mean	% or SD	Cronbach's alpha
Self-harm: HBSC measure	No (n, %)	467	73.0		451	79.7	
	Yes (n, %)	131	20.5		71	12.5	
	Missing (n, %)	42	6.6		44	7.8	
	If yes:						
	Every day/several times per week (n, %)	18	13.7		14	19.7	
	Once a week/a few times a month (n, %)	29	22.1		17	23.9	
	Once a month/several times per year (n, %)	53	40.5		31	43.7	
	Missing (n, %)	31	5.2		9	12.7	
Bullying victimisation: GBS	Total score (mean, SD)	0.4	0.5	0.66	0.3	0.5	0.72
	Missing (n, %)	120	18.8		79	14.0	
Cyberbullying: (DAPHNE II questionnaire)	No victimisation (n, %)	522	81.6		461	81.4	
	Any victimisation (n, %)	67	10.5		63	11.1	
	Missing victimisation (n, %)	51	8.0		42	7.4	
	No perpetration (n, %)	525	82.0		425	75.1	
	Any perpetration (n, %)	67	10.5		91	16.1	
	Missing perpetration (n, %)	48	7.5		50	8.8	
Substance use: smoking in last month	No (n, %)	584	91.3		496	87.6	
	Yes (n, %)	11	1.7		22	3.9	
	Missing (n, %)	45	7.0		48	8.5	
	If yes:						
	Once or twice (n, %)	1	0.2		13	2.5	
	About once a week or more (n, %)	1	0.2		5	1.0	
	Missing (n, %)	9	1.5		4	0.8	

		Year-7 (baseline) survey		Year-10 (follow-up) survey			
Indicative outcome measures		Number or mean	% or SD	Cronbach's alpha	Number or mean	% or SD	Cronbach's alpha
Substance use: drunk alcohol in last month	No (n, %)	570	89.1		467	82.5	
	Yes (n, %)	23	3.6		49	8.7	
	Missing (n, %)	47	7.3		50	8.8	
	If yes:						
	Once or twice (n, %)	12	2.0		33	6.4	
	About once a week or more (n, %)	3	0.5		11	2.1	
	Missing (n, %)	8	1.3		5	1.0	
Substance use: offered drugs	No (n, %)	549	85.8		403	71.2	
	Yes, but didn't try (n, %)	33	5.2		83	14.7	
	Yes and did try (n, %)	8	1.3		23	4.1	
	Missing (n, %)	50	7.8		57	10.1	
Substance use: cannabis	Tried in the last month (n, %)	1	0.2		13	2.6	
	Tried longer than last month (n, %)	1	0.2		4	0.8	
	Not selected (n, %)	6	1.0		6	1.2	
Substance use: glue/solvent	Tried in the last month (n, %)	2	0.3		9	1.8	
	Tried longer than last month (n, %)	2	0.3		5	1.0	
	Not selected (n, %)	4	0.7		9	1.8	
Substance use: other drugs	Tried in the last month (n, %)	4	0.7		6	1.2	
	Tried longer than last month (n, %)	0	0.0		6	1.2	
	Not selected (n, %)	4	0.7		11	2.2	
BBSCQ	Total score (mean, SD)	3.1	0.4	0.93	2.9	0.5	0.92
	Missing (n, %)	6	0.9		7	1.2	

DOI: 10.3310/GFDT2323

unique identification codes, were feasible, indicating that in a Phase III trial, it would be feasible to link baseline and follow-up data.

## Conclusion

Our study successfully piloted trial and survey operating procedures, as well as indicative primary and secondary outcome measures at baseline and follow-up to be used in a potential future trial to evaluate the effectiveness of LTMH intervention. The methods and these measures used could be feasibly scaled up to use within such a larger study. We found, broadly, that these were feasible to implement and acceptable to students, teachers and parents. With some minor amendments, these should be applied in a future Phase III effectiveness trial of the LTMH intervention. The progression criterion for trial feasibility was surpassed, in that three of four schools at baseline and two of four schools at follow-up had responses rates > 60%. Response rates in a future Phase III trial of the LTMH intervention could be improved if surveys are not scheduled in the last weeks of term. on Fridays or near mock GCSE exams. Understanding and completion rates of some survey items, such as FAS and EDE-QS, may be improved for year-7 students if question wording was more tailored to their level of reading comprehension.

## **Additional information**

## **CRediT** contribution statement

**Oliver Lloyd-Houldey (https://orcid.org/0000-0001-6081-3546):** Investigation (lead), Writing – original draft (lead), Methodology (supporting).

Joanna Sturgess (https://orcid.org/0000-0001-7312-4598): Formal analysis (lead), Data curation (lead), Methodology (supporting), Writing – reviewing and editing (supporting).

**Neisha Sundaram (https://orcid.org/0000-0003-4159-9518)**: Writing – reviewing and editing (supporting), Investigation (supporting), Methodology (supporting).

**Steven Hope (https://orcid.org/0000-0001-6759-6927)**: Methodology (supporting), Investigation (supporting), Writing – reviewing and editing (supporting).

Semina Michalopoulou (https://orcid.org/0000-0003-3198-0411): Investigation (supporting), Writing – reviewing and editing (supporting).

**Elizabeth** Allen (https://orcid.org/0000-0002-2689-6939): Formal analysis (supporting), Methodology (supporting), Writing – reviewing and editing (supporting).

Lee Hudson (https://orcid.org/0000-0003-0883-8791): Methodology (supporting), Writing – reviewing and editing (supporting).

**Stephen Scott (https://orcid.org/0000-0003-4680-6213)**: Methodology (supporting), Writing – reviewing and editing (supporting).

**Dasha Nicholls (https://orcid.org/0000-0001-7257-6605)**: Methodology (supporting), Writing – reviewing and editing (supporting).

**Deborah Christie (https://orcid.org/0000-0002-6214-2325)**: Methodology (supporting), Writing – reviewing and editing (supporting).

**Rosa Legood (https://orcid.org/0000-0003-2761-5227)**: Methodology (supporting), Writing – reviewing and editing (supporting).

**Chris Bonell (https://orcid.org/0000-0002-6253-6498)**: Conceptualisation (equal), Methodology (equal), Supervision (equal), Writing – reviewing and editing (equal).

**Russell Viner (https://orcid.org/0000-0003-3047-2247)**: Conceptualisation (equal), Methodology (equal), Supervision (equal), Writing – reviewing and editing (equal).

#### Data-sharing statement

All data requests should be submitted to the corresponding author for consideration. Access to all anonymised data may be granted following review.

#### **Ethics statement**

Ethics approval for the Learning Together for Mental Health study UCL Research Ethics Committee (REC) on 30 March 2022 (UCL Ethics Project ID Number: 21179/001) and the London School of Hygiene & Tropical Medicine REC 26 August 2022 (ref. 27994).

#### Information governance statement

University College London and the London School of Hygiene and Tropical Medicine are committed to handling all personal information in line with the UK Data Protection Act (2018) and the General Data Protection Regulation (EU GDPR) 2016/679. Under Data Protection legislation University College London is the Data Controller, and we process personal data in accordance with their instructions. The UCL Data Protection Officer provides

14

oversight of UCL activities involving the processing of personal data, and can be contacted at data-protection@ucl.ac.uk. Further information on how UCL uses participant information can be found in its 'general' privacy notice 'for participants in health and care research studies' here: www.ucl.ac.uk/legal-services/ privacy/participants-health-and-care-research-privacy-notice.

#### Disclosure of interest statement

*Full disclosure of interests:* Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/GFDT2323.

**Primary conflicts of interest:** Rosa Legood, Neisha Sundaram, Oliver Lloyd-Houldey, Joanna Sturgess, Elizabeth Allen and Russell Viner are funded by several NIHR grants. Dasha Ncholls is supported by the NIHR Applied Research Collaboration Northwest London and NIHR Imperial BRC. Russell Viner is Chief Science Advisor at the Department for Education. Chris Bonell and Russell Viner are NIHR Senior Investigators. Chris Bonell is a member of the NIHR Reduce Demand for Illicit Substances funding board. The views expressed are those of the authors and not necessarily those of the NIHR or the Department of Health and Social Care. All authors declare they have no competing interests.

# Department of Health and Social Care disclaimer

This publication presents independent research commissioned by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, MRC, NIHR Coordinating Centre, the Public Health Research programme or the Department of Health and Social Care.

This article was published based on current knowledge at the time and date of publication. NIHR is committed to being inclusive and will continually monitor best practice and guidance in relation to terminology and language to ensure that we remain relevant to our stakeholders.

#### Study registration

This study is registered as Current Controlled Trials ISRCTN15301591.

#### Funding

This article presents independent research funded by the National Institute for Health and Care Research (NIHR) Public Health Research programme as award number NIHR131594.

This article reports on one component of the research award Learning Together to promote mental health and wellbeing (LTMH). For other articles from this thread and for more information about this research, please view the award page (www.fundingawards.nihr. ac.uk/award/NIHR131594).

#### About this article

The contractual start date for this research was in January 2021. This article began editorial review in August 2024 and was accepted for publication in January 2025. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The Public Health Research editors and publisher have tried to ensure the accuracy of the authors' article and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this article.

#### Copyright

Copyright © 2025 Lloyd-Houldey *et al.* This work was produced by Lloyd-Houldey *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/ licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

## List of supplementary material

**Report Supplementary Material 1** Baseline year-7 student questionnaire

Supplementary material can be found on the NIHR Journals Library article page (https://doi. org/10.3310/GFDT2323).

Supplementary material has been provided by the authors to support the report, and any files provided at submission will have been seen by peer reviewers, but not extensively reviewed. Any supplementary material provided at a later stage in the process may not have been peer reviewed.

## List of abbreviations

BBSCQ	Beyond Blue School Climate Questionnaire
DBS	Disclosure and Barring Services
EDE-QS	Eating Disorders Examination – Questionnaire Short
FAS	Family Affluence Scale
FSM	free school meal
GAD-7	Generalised Anxiety Disorder-7
GBS	Gatehouse Bullying Scale
GCSE	General Certificate of Secondary Education
HBSC	Health Behaviour in School-aged Children
LSHTM	London School of Hygiene and Topical Medicine
LT	Learning Together
LTMH	Learning Together for Mental Health
MH	mental health
MRC	Medical Research Council
PPIE	patient and public involvement and engagement
REC	Research Ethics Committee
RP	restorative practice
SAE	serious adverse event
SDQ	Strengths and Difficulties Questionnaire
SEL	social and emotional learning
SMFQ	Short Moods and Feelings Questionnaire
SOP	standard operating procedure
SWEMWBS	Short Warwick-Edinburgh Mental Well-being Scale
UCL	University College London

## References

16

- 1. NHS England Mental Health Taskforce. *The Five Year Forward View for Mental Health*. London: NHS England; 2016.
- 2. Solmi M, Radua J, Olivola M, Croce E, Soardo L, Salazar de Pablo G, *et al.* Age at onset of mental disorders

worldwide: large-scale meta-analysis of 192 epidemiological studies. *Mol Psychiatry* 2022;**27**:281–95. https://doi.org/10.1038/s41380-021-01161-7

- Kessler RC, Berglund P, Demler O, Jin R, Merikangas KR, Walters EE. Lifetime prevalence and age-of-onset distributions of DSM-IV disorders in the National Comorbidity Survey Replication. Arch Gen Psychiatry 2005;62:593-602. https://doi.org/10.1001/archpsyc. 62.6.593
- 4. Newlove-Delgado T, Marcheselli F, Williams T, Mandalia D, Davis J, McManus S, et al. Mental Health of Children and Young People in England, 2023. Leeds: NHS England; 2023.
- Bould H, De Stavola B, Lewis G, Micali N. Do disordered eating behaviours in girls vary by school characteristics? A UK cohort study. *Eur Child Adolesc Psychiatry* 2018;27:1473–81. https://doi. org/10.1007/s00787-018-1133-0
- 6. McManus S, Bebbington P, Jenkins R, Brugha T. Mental Health and Wellbeing in England: Adult Psychiatric Morbidity Survey 2014. Leeds: NHS Digital; 2016.
- 7. Transforming Children and Young People's Mental Health Provision: A Green Paper. London: Department of Health and Department for Education, England; 2018.
- NHS England. Mental Health Support in Schools and Colleges. NHS England. URL: www.england.nhs.uk/ mental-health/cyp/trailblazers/ (accessed 30 January 2024).
- O'Reilly M, Svirydzenka N, Adams S, Dogra N. Review of mental health promotion interventions in schools. Soc Psychiatry Psychiatr Epidemiol 2018;53:647-62. https://doi.org/10.1007/s00127-018-1530-1
- Pearcey S, Shum A, Waite P, Patalay P, Creswell C. Co-Space Study. Report 04: Changes in children and young people's emotional and behavioural difficulties through lockdown, 2020. Co-SPACE study. URL: https://cospaceoxford.org/findings/4th-update/ (accessed 30 January 2024).
- 11. Levita L, Miller JG, Hartman TK, Murphy J, Shevlin M, McBride O, *et al.* Report2. Impact of Covid-19 on young people aged 13-24 in the UK- preliminary findings 2020. https://doi.org/10.31234/osf.io/s32j8
- Caldwell DM, Davies SR, Hetrick SE, Palmer JC, Caro P, López-López JA, *et al.* School-based interventions to prevent anxiety and depression in children and young people: a systematic review and network metaanalysis. *Lancet Psychiatry* 2019;6:1011–20. https:// doi.org/10.1016/S2215-0366(19)30403-1
- Neil AL, Christensen H. Efficacy and effectiveness of school-based prevention and early intervention programs for anxiety. *Clin Psychol Rev* 2009;29:208–15. https://doi.org/10.1016/j.cpr.2009.01.002

- Sanchez AL, Cornacchio D, Poznanski B, Golik AM, Chou T, Comer JS. The Effectiveness of School-Based Mental Health Services for Elementary-Aged Children: a meta-analysis. J Am Acad Child Adolesc Psychiatry 2018;57:153–65. https://doi.org/10.1016/j. jaac.2017.11.022
- 15. Paulus FW, Ohmann S, Popow C. Practitioner review: school-based interventions in child mental health. J Child Psychol Psychiatry 2016;**57**:1337–59. https:// doi.org/10.1111/jcpp.12584
- Yager Z, Diedrichs PC, Ricciardelli LA, Halliwell E. What works in secondary schools? A systematic review of classroom-based body image programs. *Body Image* 2013;10:271–81. https://doi.org/10.1016/j. bodyim.2013.04.001
- Pandey A, Hale D, Das S, Goddings AL, Blakemore SJ, Viner RM. Effectiveness of universal selfregulation-based interventions in children and adolescents: a systematic review and meta-analysis. JAMA Pediatr 2018;172:566-75. https://doi.org/10.1001/ jamapediatrics.2018.0232
- Stallard P, Skryabina E, Taylor G, Phillips R, Daniels H, Anderson R, Simpson N. Classroom-based cognitive behaviour therapy (FRIENDS): a cluster randomised controlled trial to Prevent Anxiety in Children through Education in Schools (PACES). *Lancet Psychiatry* 2014;1:185–92. https://doi.org/10.1016/ S2215-0366(14)70244-5
- 19. Herlitz L, MacIntyre H, Osborn T, Bonell C. The sustainability of public health interventions in schools: a systematic review. *Implement Sci* 2020;**15**:4. https:// doi.org/10.1186/s13012-019-0961-8
- Langford R, Bonell C, Jones H, Pouliou T, Murphy S, Waters E, *et al.* The World Health Organization's Health Promoting Schools framework: a Cochrane systematic review and meta-analysis. *BMC Public Health* 2015;**15**:130. https://doi.org/10.1186/ s12889-015-1360-y
- 21. Bonell C, Jamal F, Harden A, Wells H, Parry W, Fletcher A, *et al.* Systematic review of the effects of schools and school environment interventions on health: evidence mapping and synthesis. *Public Health Res* 2013;**1**:1–320. https://doi.org/10.3310/phr01010
- 22. Farahmand FK, Grant KE, Polo AJ, Duffy SN, DuBois DL. School-based mental health and behavioral programs for low-income, urban youth: a systematic and metaanalytic review. *Clin Psychol Sci Pract* 2011;**18**:372–90. https://doi.org/10.1111/j.1468-2850.2011.01265.x
- 23. Bonell C, Beaumont E, Dodd M, Elbourne DR, Bevilacqua L, Mathiot A, *et al.* Effects of school environments on student risk-behaviours: evidence from a longitudinal study of secondary schools in England.

J Epidemiol Community Health 2019;73:502–8. https:// doi.org/10.1136/jech-2018-211866

- 24. Bonell C, Allen E, Opondo C, Warren E, Elbourne DR, Sturgess J, et al. Examining intervention mechanisms of action using mediation analysis within a randomised trial of a whole-school health intervention. J Epidemiol Community Health 2019;73:455–64. https:// doi.org/10.1136/jech-2018-211443
- 25. Baker DP, Leon J, Smith Greenaway EG, Collins J, Movit M. The education effect on population health: a reassessment. *Popul Dev Rev* 2011;**37**:307–32. https:// doi.org/10.1111/j.1728-4457.2011.00412.x
- Bonell C, Wells H, Harden A, Jamal F, Fletcher A, Thomas J, et al. The effects on student health of interventions modifying the school environment: systematic review. J Epidemiol Community Health 2013;67:677-81. https://doi.org/10.1136/jech-2012-202247
- Leurent B, Dodd M, Allen E, Viner R, Scott S, Bonell C. Is positive school climate associated with better adolescent mental health? Longitudinal study of young people in England. SSM Ment Health 2021;1:100033. https://doi.org/10.1016/j.ssmmh.2021.100033
- 28. Humphrey N. Are the kids alright? Examining the intersection between education and mental health. *Psychol Educ Rev* 2018;**42**:4–16.
- Bonell C, Allen E, Warren E, McGowan J, Bevilacqua L, Jamal F, et al. Effects of the Learning Together intervention on bullying and aggression in English secondary schools (INCLUSIVE): a cluster randomised controlled trial. Lancet (London, England) 2018;392:2452-64. https://doi.org/10.1016/S0140-6736(18)31782-3
- Bonell C, Hope S, Sundaram N, Lloyd-Houldey O, Michalopoulou S, Scott S, *et al.* Public engagement to refine a whole-school intervention to promote adolescent mental health [published online ahead of print December 4 2024]. *Public Health Res* 2024. https:// doi.org/10.3310/JWGT4863
- 31. Lordan G, McGuire A. Healthy Minds: Health Outcomes Evaluation Report and Executive Summary. London: Education Endowment Foundation; 2019.
- Coleman J, Hale D, Layard R. A Model for the Delivery of Evidence-Based PSHE (Personal Wellbeing) in Secondary Schools. CEP Discussion Paper No. 1071. August 2011. Centre for Economic Performance LSE; 2011. URL: https://cep.lse.ac.uk/pubs/download/dp1071. pdf (accessed 24 January 2024).
- 33. Sundaram N, Lloyd-Houldey O, Michalopoulou S, Hope S, Sturgess J, Allen E, *et al.* Qualitative study of the feasibility and acceptability of implementation, and potential mechanisms of Learning Together for Mental Health, a whole-school intervention aiming to promote mental health and wellbeing in secondary

This article should be referenced as follows:

Lloyd-Houldey O, Sturgess J, Sundaram N, Hope S, Michalopoulou S, Allen E, et al. Learning together for mental health: feasibility of measures to assess a whole-school mental health and wellbeing intervention in secondary schools [published online ahead of print June 25 2025]. Public Health Res 2025. https://doi.org/10.3310/GFDT2323

schools. Pilot Feasibility Stud 2024;10:142. https://doi. org/10.1186/s40814-024-01563-8

- 34. Sundaram N, Lloyd-Houldey O, Michalopoulou S, Hope S, Sturgess J, Allen E, et al. Qualitative study of the feasibility and acceptability of implementation, and potential mechanisms of learning together for mental health, a whole-school intervention aiming to promote mental health and wellbeing in secondary schools. *Pilot Feasibility Stud*. 2024 Nov 15;10:142. https://doi.org/10.1186/s40814-024-01563-8
- 35. Goodman R, Ford T, Simmons H, Gatward R, Meltzer H. Using the Strengths and Difficulties Questionnaire (SDQ) to screen for child psychiatric disorders in a community sample. *Br J Psychiatry* 2000;**177**:534–9. https://doi.org/10.1192/bjp.177.6.534
- 36. Sadler K, Vizard T, Ford T, Goodman A, Goodman R, McManus S. Mental Health of Children and Young People in England, 2017: Summary of Key Findings. London: NHS Digital; 2018.
- 37. Newlove-Delgado T, Marcheselli F, Williams T, Mandalia D, Davis J, McManus S. Mental Health of Children and Young People in England, 2022. Leeds: NHS Digitall; 2022.
- 38. Kuyken W, Nuthall E, Byford S, Crane C, Dalgleish T, Ford T, *et al.*; MYRIAD team. The effectiveness and cost-effectiveness of a mindfulness training programme in schools compared with normal school provision (MYRIAD): study protocol for a randomised controlled trial. *Trials* 2017;**18**:194. https://doi.org/10.1186/s13063-017-1917-4
- 39. Deighton J, Lereya ST, Morgan E, Breedvelt J, Martin K, Feltham A, et al. Measuring and Monitoring Children and Young People's Mental Wellbeing: A Toolkit for Schools and Colleges. London: Public Health England and the UCL Anna Freud Evidence Based Practice Unit; 2016.
- Spitzer RL, Kroenke K, Williams JBW, Löwe B. A Brief Measure for Assessing Generalized Anxiety Disorder: The GAD-7. Arch Intern Med 2006;166:1092–7. https://doi.org/10.1001/archinte.166.10.1092
- 41. Engelsen BK, Laberg JC. A comparison of three questionnaires (EAT-12, EDI, and EDE-Q) for assessment of eating problems in healthy female adolescents. *Nord J Psychiatry* 2001;**55**:129–35.
- 42. Brooks F, Klemera E, Chester K, Magnusson J, Spencer N. HBSC England National Report: Findings from the 2018 HBSC Study for England. Hatfield, England: University of Hertfordshire; 2020.
- 43. Bond L, Wolfe S, Tollit M, Butler H, Patton G. A comparison of the gatehouse bullying scale and the

peer relations questionnaire for students in secondary school. *J Sch Health* 2007;**77**:75–9. https://doi. org/10.1111/j.1746-1561.2007.00170.x

- 44. Smith PK, Mahdavi J, Carvalho M, Fisher S, Russell S, Tippett N. Cyberbullying: its nature and impact in secondary school pupils. *J Child Psychol Psychiatry* 2008;**49**:376–85. https://doi. org/10.1111/j.1469-7610.2007.01846.x
- 45. Bridges S, Gill V, Omole T, Sutton R, Wright V. Smoking, Drinking and Drug Use Among Young People in England in 2010. London: NHS Information Centre for Health and Social Care; 2011.
- 46. Sawyer MG, Pfeiffer S, Spence SH, Bond L, Graetz B, Kay D. School-based prevention of depression: a randomised controlled study of the beyondblue schools research initiative. J Child Psychol Psychiatry 2010;51:199–209. https://doi. org/10.1111/j.1469-7610.2009.02136.x
- 47. Medical Research Council. Medical Research Council (MRC) Ethics Guide. Medical Research Involving Children. London: MRC; 2004.
- 48. Children's Legal Centre. Gillick v West Norfolk and Wisbech Area Health Authority and Department of Health and Social Security [1984] Q.B. 581. As cited in Children's Legal Centre (1985) Landmark decision for children's rights. Childright 1985;22:11–8.
- 49. Ponsford R, Bragg S, Allen E, Tilouche N, Meiksin R, Emmerson L, et al. A school-based social-marketing intervention to promote sexual health in English secondary schools: the Positive Choices pilot cluster RCT. Public Health Res 2021;9:1–190. https://doi. org/10.3310/phr09010
- 50. FFT Education Datalab. Weekday Attendance Analysis: A New Report for Schools. Online: FFT Education Datalab; 2022 [cited 2022 6 December]. URL: https:// ffteducationdatalab.org.uk/2022/12/weekday-attendance-analysis-a-new-report-for-schools/ (accessed 12 December 2023).
- 51. Beynon K, Thomson D. Absence from School on Fridays. FFT Education Datalab; 2023 [cited 2023 8 March]. URL: https://ffteducationdatalab.org.uk/2023/03/ absence-from-school-on-fridays/#:~:text=And%20 we%20split%20the%20data,to%20around%20 3.5%25%20of%20pupils (accessed 12 December 2023).
- 52. Long R, Roberts N. School Attendance in England. London: House of Commons Library; 2024.